Author’s response to reviews

Title: Justification for the use of Ocimum gratissimum L in herbal medicine and its interaction with disc antibiotics.

Authors:

Emeka I Nweze (nwezemeka@yahoo.com)
Elizabeth E Eze (mezechoil@yahoo.com)

Version: 3 Date: 23 September 2009

Author’s response to reviews:

Response to Reviewers comments No2.

1. 1) the new table 3: third line from the bottom. The value 7.14 has to be revised.
   May be – 7.14??
   Answer: Yes we agree. The value is -7.14 and has been corrected in the manuscript.

2. 2) spelling mistakes remain again twice the world anthroquinone-----> anthraquinone p7 and p14.
   Answer: The corrected spellings have been added. It was an over sight.

3. 3) The sentence « Conclusively, the result of the interaction of the leaves of the plant with the... » has to be corrected as « « Conclusively, the result of the interaction of the leaves extract of the plant with the ... ».
   Answer: The phrase ‘of the leaves’ was repeated twice and we have removed one of that.

We thank the Reviewer for the job well done and look forward to the final publication of the manuscript. Thanks a lot